Author's response to reviews

Title: Benign teratoma of the nasal septum: a case report

Authors:

Ibrahim CUKUROVA (cukurova57@gmail.com)
Murat GUMUSSOY (mgumussoy@hotmail.com)
Aytekin YAZ (aytekinyaz@hotmail.com)
Umit BAYOL (umitbayol@yahoo.com)
Orhan G YIGITBASI (gaziyigitbasi@yahoo.com)

Version: 2 Date: 19 January 2012

Author's response to reviews: see over
Michael Kidd AM,
*Editor – in – Chief*
*Flinders University,*
*Australia*

January 18, 2012

Dear Michael Kidd AM,

On behalf of my co-authors, I would like to submit the enclosed manuscript entitled “Benign teratoma of the nasal septum: a case report” for publication in *Journal of Medical Case Reports.* It has not been submitted for publication nor has it been published in whole or in part elsewhere. It will not be published elsewhere without permission of the Editors. The language of the manuscript was edited by a native speaker. I attest to the fact that all authors listed on the title page have read the manuscript, attest to the validity and legitimacy of the data and its interpretation, and agree to its submission.

Since our study presents a disease that is rarely observed in elderly patients, we think that it should be published in the journal.

The study demonstrates an original case that is rarely observed in elderly patients.

I ensure that there is no conflict of interest of the authors.

I would like to suggest following reviewers for the evaluation of the manuscript:

1. Cemal CİNGİ, M.D., Prof.
   E-mail: ccingi@gmail.com
   Institution: Eskişehir Osmangazi University Faculty of Medicine, Department of Ear-Nose-Throat, Eskişehir, TURKEY

2. Hamdi ÇAKLI, M.D., Assoc. Prof.
   E-mail: hcakli@ogu.edu.tr
I kindly look forward for the result of review of manuscript.

Kindest regards,

**Corresponding Author**

**Name:** Murat GÜMÜŞSOY

**Address:** Department of Otolaryngology and Head and Neck Surgery, the Ministry of Health
Izmir Tepecik Training and Research Hospital, Izmir, 35120, Turkey

**Phone:** +90 232 469 69 69/16 16

**Fax:** +90 232 433 07 56

**E-mail:** mgumussoy@hotmail.com